Author's response to reviews

Title: Surgeons agree more on treatment recommendations than on classification of proximal humeral fractures

Authors:

Stig Brorson (sbroson@hotmail.com)
Bo S Olsen (bosaol01@heh.regionh.dk)
Lars H Frich (frichs@damlnd.net.dk)
Steen L Jensen (steen.lund.jensen@rn.dk)
Anne Kathrine B Sørensen (ankaso01@heh.regionh.dk)
Michael R Krogsgaard (mrk@damlnd.net.dk)
Asbjørn Hróbjartsson (ah@cochrane.dk)

Version: 3  Date: 12 May 2012

Author's response to reviews:

Dear Editor,

Please find attached our manuscript MS: 2385097326590674 ‘Surgeons agree more on treatment recommendations than on classification of proximal humeral fractures# submitted 12 January 2012.

The manuscript was returned 10 May 2012 and we were asked to clarify why ethical approval was not required for our study. Please find our comments below:

We included an historical series of blinded radiographs from five institutions presented in a random order. No images could be linked to any patient, surgeon, or institution.

Please let me know if you would like to check our files or make any changes in our manuscript to comply with BMC Musculoskeletal Disorders’ house style.

Thank you for considering our manuscript for publication in BMC Musculoskeletal Disorders.

Sincerely,
Stig Brorson, MD PhD